# Measuring the effects of the tipped minimum wage using W-2 data

Maggie R. Jones\*

U.S. Census Bureau

June 23, 2015

#### Abstract

Researchers have found it difficult to capture the hourly wages of tipped workers and thus assess the economic effects of the tipped minimum wage. In this paper, I present a new measure of hourly wages for tipped servers (wait staff and bartenders) using linked W-2 and survey data. I also estimate the effect of tipped minimum wages on wages, hourly tips, server employment, and hours worked. To provide an assessment of the quality of the data, I compare the W-2 measures to comparable responses in the survey data. I find that higher tipped minimum wages raise that portion of wages paid by employers, but decrease tip income by a similar percentage. I also find evidence of a quadratic relationship between tipped minimum wages, where employment first increases and then decreases in response to higher wages. The results are consistent with a monopsony model of server employment.

Keywords: Minimum wages, restaurant industry, monopsony

JEL classification: J21, J23, J30, J31, J38

\*The views expressed in this paper are those of the author and do not necessarily represent the views of the U.S. Census Bureau.

### 1 Introduction

While the effects of the minimum wage on the wages and employment of low-income workers have received a great deal of attention in the economic literature, the effects of tipped minimum wages have received little attention. Even those minimum wage studies that have focused on the restaurant industry, which employs a large share of employees who are covered by the tipped minimum wage, have not separately analyzed it. Rather, recognizing that the restaurant industry has a large number of minimum wage workers who are not tipped, these analyses have pooled all restaurant workers and assumed that a single wage floor applies to them. Or, they have focused only on the limited-service industry, in which workers are not tipped.

The wage floors that apply to tipped and non-tipped workers may be very different from one another on a state-by-state basis. The wage that is paid to a tipped server directly by his or her employer is usually much less than the employer must pay to a cook or dishwasher (at the federal level, \$2.13 versus \$7.25 as of 2014). Moreover, while there is some correlation between state minimum wages and tipped minimum wages, the difference between the two is often quite large. For example, in Massachusetts the non-tipped minimum wage was \$8 and the tipped minimum wage was \$2.63 for 2011.

The main source of information on how much tipped earners receive per hour is the Current Population Survey Outgoing Rotation Group (CPS ORG). Besides the usual information on industry and occupation, the survey asks respondents whether they receive tips as part of their hourly compensation and to report on what they make per hour minus tips (a second question asks about weekly earnings including tips). The survey response suffers from significant error (Rodgers et al., 1993), with respondents tending to overestimate at the lower end of the distribution and underestimate at the top end. Several researchers have coped with the problematic nature of survey responses by using establishment-level data such as the Quarterly Census of Employment and Wages; however, it is impossible with these aggregate data to differentiate between employer-

paid wages and tips. Without a reliable measure of the hourly wage paid by employers, the effect of tipped minimum wages on the welfare of servers is difficult to assess.

Yet the topic of tipped minimum wages is an important one for a variety of reasons. Recently, there has been increased focus on the federal tipped minimum, which has not been increased since 1996.<sup>1</sup> Discussions of minimum wage increases are always contentious, and the importance of good data and analysis on the topic cannot be overestimated. This is especially true because of the number of tipped employees in the workforce. For example, fully 2.2 million workers were employed in service industries in 2013, representing three-fifths of all workers earning at or less than the federal minimum wage. Moreover, the recent economic recovery has seen a disproportionate growth in the food and beverage sector, with the "food services and drinking places industry [accounting] for almost 1 out of every 6 nonfarm jobs added during the recovery," according to the Bureau of Labor Statistics. <sup>2</sup> The expectation is that an even higher number of workers will be covered by tipped minimum wages in future years.

In this paper, I use a unique data set that overcomes the problems of survey and establishment-level data—linked administrative and survey records that permit the calculation of an hourly wage for both employer-paid wages and tips for tipped servers in the restaurant industry. Using W-2 data from the Internal Revenue Service (IRS) linked with Current Population Survey Annual Social and Economic Supplement (CPS ASEC) data for 2005–2011, I create wage equivalents that capture hourly wages paid by employers (and thus subject to tipped minimum wages) and hourly tips. I discuss the pros and cons of these new measures and compare them to the measures in the CPS ORG for a subset of the sample. I then examine the effects of tipped minimum wages on hourly wages, hourly tips, server employment, and hours worked per year, using the variation in tipped minimum wages in states over time as an identification strategy.

<sup>&</sup>lt;sup>1</sup>Much of the focus at the moment is due to a bill being considered in the House of Representatives: http://democrats.edworkforce.house.gov/issue/fair-minimum-wage-act.

<sup>&</sup>lt;sup>2</sup>http://www.bls.gov/iag/tgs/iag722.htm

I find that hourly wages paid by employers increased as tipped minimums increased, with an elasticity of 0.5 to 0.7, and that hourly tips decreased by a similar amount. I also find some evidence that employment first increases and then falls in response to tipped minimum wages. These results are consistent with a monopsony model of server employment. My results contribute to the literature on tipped minimum wages specifically, and minimum wages in general, in several ways. First, my measures are precise enough to show how wages and tips may cancel one another out, leaving a researcher using a total wage measure to find no effect. Second, the monopsony effect of first increasing and then decreasing employment provides some policy guidance on how high to set a tipped minimum wages if we suspect that the labor market in general follows a monopsony model. Finally, while the results show that servers do not ultimately make more in total hourly wage, for most of the range of the tipped minimum they benefit from higher employment levels and, thus, work less for their tips.

The paper proceeds as follows: Section 2 provides background on the tip credit and an overview of the relevant literature. Section 3 goes over the data, with particular emphasis on how W-2 data is generated, the reporting requirements of employers and servers, and the comparison of W-2 data to survey data. Section 4 provides some information on the usual theories regarding tipped servers and the minimum wage, and gives the empirical specification used in the analysis. Section 5 presents the results, and Section 6 concludes. Also included is an appendix providing some detail on the W-2 and the reporting requirements for employers of tipped employees, as well as a stateby-state accounting of wage information.

### 2 Background

#### 2.1 Tip credit policies

The Fair Labor Standards Act of 1938 established the "tip credit" for employers whose workers generally receive a substantial proportion of their income in tips. Originally, the credit was set in such a way that it made up no more than 50 percent of the standard hourly minimum wage. In 1996, an amendment to the law decoupled the tipped minimum from the standard minimum, and since that time the federal tipped minimum wage has remained at \$2.13 per hour. This rate reflects the standard minimum wage in place in 1996 (\$4.25 per hour). Thus, as the standard minimum wage has increased over time, so has the allowable tip credit, which currently stands at \$5.12 per hour (Robinson, 2011).

States have responded to the lack of federal changes in the tipped minimum wage by altering the tip credit. At the extreme end, seven states (as of 2013) have eliminated the tip credit entirely, requiring employers to pay the standard federal or state minimum wage to employees regardless of whether they also receive tips. Tip credits in other states range from very close to the federal tip credit (Delaware) to an almost zero tip credit (Hawaii).<sup>3</sup> Figure 1 shows maps of the U.S. in 2005 and 2011, giving an indication of the variation in tipped minimum wages and how they changed over the time period. The categories shown in the maps are the federal tipped minimum (\$2.13), and two, three, and (for 2011) four times that amount. Compared with standard state minimum wages, which ranged from \$7.25 to \$8.75 as of 2015, tipped minimums have a wider variation. Between 2005 and 2011, there were 85 separate state changes to tipped minimums, with enough variation in the measure over time and place to identify an effect.

The regulations regarding the tip credit are outlined in the federal code. To apply the

<sup>&</sup>lt;sup>3</sup>Allegretto (2013) provides an overview of the tipped minimum wage compared with the standard minimum wage

credit to an employee's hourly wage, the employee must regularly receive more than \$30 in tips in a month, must be informed of the credit and how it is applied, and must be permitted to retain all of the tips unless there is an appropriate tip-pooling arrangement.<sup>4</sup> Federal regulations also require careful recording of tipped and non-tipped work by an individual employee.<sup>5</sup> Both employers and employees have the responsibility to record tip income for tax purposes. This includes recordkeeping through point-of-sale (POS) systems that can keep track of tips written on credit card slips, as well as tip logs showing the cash tips received during a shift. Historically, the IRS has pursued legal action against restaurant owners in cases of income misreporting, as opposed to employees, as it is less costly to act against a single entity rather than multiple individuals (Peckron, 2002). Thus, employers have a strong incentive to accurately record the tips made by their employees.

The tax regulations regarding the recording of tips center on the payment of payroll taxes, for which tips are treated as regular wages. The record of the wages that are paid by an employer to a tipped employee are recorded separately on the W-2; the wages made in tips, and the payroll taxes paid on those tips, are then recorded in separate fields. All wages—from the employer and from tips—are then recorded in the familiar "Wages, Salary, and Tips" field of the W-2.

#### 2.2 Literature review

There is an extensive literature on the impacts of minimum wages, but little research has been done specifically on the tipped minimum wage. A likely reason for this gap in the literature is the absence of reliable data on tips. The ORG data from the CPS is the

<sup>&</sup>lt;sup>4</sup>Tip-pooling regulations vary slightly by state, but generally allow employers to require tipped employees to contribute a certain percentage of their tips into a pool, as long as the contributed amount does not lower an employees hourly wage to less than the minimum wage. No portion of pooled tips may go to the employer or to employees who generally don't receive tips (e.g., cooks and dishwashers).

<sup>&</sup>lt;sup>5</sup>This is to keep employers from assigning duties to tipped employees that would otherwise be performed by a non-tipped worker paid at the standard minimum.

main source of information on a variety of employment measures—specifically hourly wages. Research using administrative records linked with survey data has established that the reporting of hourly wages suffers from error (Moore et al., 2000; Rodgers et al., 1993). Yet it seems likely that most employees who are paid an hourly wage set by an employer know, more or less, what they receive per hour. The same cannot be said for tipped employees, whose hourly wage will vary depending on the tips received. Wait staff who work a shift on a slow day will make much less per hour than they will during a busy dinner shift. Wages will also be dependent on the season in many geographic areas. A typical server will likely have difficulty calculating an average rate of pay over a given period.

Brown (1999) provides a thorough overview of minimum wage research up to the late 1990s; Neumark and Wascher (2006) gives an overview of the "new minimum wage research" since that time. The general findings of early research pointed to a negative effect of minimum wages on employment, with later modeling that used state variation in minimum wages and difference-in-differences models showing either no effect or a small positive effect. The empirical findings of these later studies fit in with models of the labor market that depart from the competitive model, such as monopsony (Boal and Ransom, 1997). A monopsony model is particularly well suited to the situation of tipped workers since the wage is expressed as a function of the number of employees, with the supply curve of labor upward sloping. Lately, minimum wage research has focused on econometrically controlling for spatial heterogeneity in panel data designs (see, for example, Dube et al. (2010)), although use of this method has been questioned (Neumark et al., 2013).

In terms of tipped wages, the earliest empirical work on the topic is Wessels (1997) (plus an earlier paper, Wessels (1993)), who developed a monopsony model of restaurant employment and analyzed the Census of Retail Trade. Without more detailed information on employees, such as occupation or number of hours worked, Wessels uses

the service time per meal (restaurant employment divided by real restaurant sales) as the dependent variable. He acknowledges that investigating the tipped wage using this measure is not ideal, as both tipped and non-tipped workers will be included in the numerator. Wessels finds evidence for the monopsony model in that employment first rises and then falls as the tipped minimum wage increases.

Since Wessels' papers, there have been a small number of analyses using survey data. Anderson and Bodvarsson (2005) investigated Occupational Employment Statistics to discover whether sub-minimum wages have "bite," finding that servers in higherwage states do not appear to see an increase in wage. Allegretto (2013) follows a strategy similar to Dube et al. (2010) to study the effects of the tipped minimum wage in the full-service restaurant industry, finding that wages and employment both increase as a response to a higher wage floor. This is in contrast to a similar paper by Even and Macpherson (2014), which estimates a positive earnings effect and a negative employment effect using the same data (the Quarterly Census of Earnings and Wages), albeit over a shorter time period.

There are key missing elements in each of these analyses. First is the separation of full hourly wage and its two parts: wage paid by the employer and tips. An employer's response to an increased minimum will affect tip income in a manner that is ambiguous, since changes in price, hours of operation, and number of employees will affect each employee's tips. However, there should be an unambiguous increase in hourly wage paid by an employer if his wages were below this threshold. Second, if at all possible it is important to separate tipped from non-tipped employees. Simply looking at sector employment conflates two separate labor markets, each of which faces a different wage floor.<sup>6</sup> I am able to contribute to the existing literature by examining measures on wages paid by employers separate from tip income through administrative records. By linking these records to survey data, I am also able to look exclusively at restaurant servers.

<sup>&</sup>lt;sup>6</sup>Oddly, there is only a weak association between state minimum wages and state tipped minimum wages—about 0.4.

### 3 Data, Sample, and Descriptives

#### 3.1 W-2 data

To my knowledge, this is the first paper to examine tipped income using W-2 administrative records. As such, a description of the data and some documentation of how it is generated is in order. The IRS directs restaurant owners to keep track of employees' wages and tips. Before the widespread use of POS systems to record this information, servers would, ideally, record tip information in a tip log and report it to their employers on a regular basis (once per shift, per week, or per month). Modern POS systems can be configured to require servers to report all tips every shift, regardless of whether they were made as cash or electronically, and the closeout of business for the day includes the tallying of all tip income (Williams, 2014). The employer is then responsible for submitting payroll taxes for his or her employees based on the tip data. For each employee, the employer must report wages and tips separately on the W-2 form.<sup>7</sup> There is also a field on the W-2 for allocated tips. Allocated tips are reported when the total tips recorded by restaurant employees for the year sum to less than eight percent of food and beverage sales. Taxes on allocated tips are paid by the employer.

The question then arises: How accurate is tip reporting? In an era when most transactions in restaurants are performed electronically, and POS systems can track all transactions performed by individual servers, it is safe to say that W-2 data is more accurate than survey responses. Like all wage data, it is unlikely to be perfect. Some tipping is still performed in cash, and much of this will go unreported by servers. Restauranteurs often employ work under the table, and such work will certainly not be recorded in tax data.<sup>8</sup> A final issue is that restaurant owners and managers may poorly understand the

<sup>&</sup>lt;sup>7</sup>See Appendix Figure 1 for W-2 instructions. The instructions specific to reporting tips are highlighted. <sup>8</sup>This aspect of hourly wage mis-measurement is unrelated to tipping per se, since other occupations suffer from under-the-table work (construction, for example). However, many restaurants owners are tempted by the off-the-books nature of tips, employing unskilled and possibly undocumented workers for tips only (ROC United, 2011).

W-2 instructions, in which case they may include tips in the FICA wage field.

On the other hand, a specific high-profile case from 2002 has put employers on their guard when it comes to tip income. In *United States* v. *Fior d'Italia*, the Supreme Court determined that a San Francisco restaurant was liable to the IRS for tip income that had gone unreported by its employees. The restaurant was required to pay \$23,000 in back FICA taxes, representing both the restaurant's and the employees' share (Peckron, 2002). Since this landmark case, employers have had a strong incentive to carefully track their employees' tip earnings, and the prevalence of debit/credit cards and POS systems has made it easy for them to do so. As technology has improved, the reporting of tips has improved (Williams, 2014).

Anecdotal evidence confirms this. In discussions with restaurant managers in the Washington, DC, area, I found that POS systems which track tips are the standard in the restaurant industry. On the other hand, some restaurants appear to pay their servers' tips in their paychecks rather than tipping out at the end of a shift. This practice may make the reporting of tipped wages even more difficult for survey respondents. It may also confound tip and wage reporting on the W-2.

#### 3.2 Survey data and variables

The survey data used in this work is the 2006 to 2012 CPS ASEC, linked person by person with W-2 data. Because the CPS ASEC captures information for the preceding tax year, each CPS file is linked to the W-2 file for the year before (for example, the 2006 CPS ASEC is linked to the 2005 W-2 data). All years are appended together, for seven years of data. Over this time the federal tipped wage remained at \$2.13 per hour, but states' tipped minimums underwent many changes.

Records were linked in the Center for Administrative Records and Research (CARRA) at the U.S. Census Bureau. The linking process involves assigning to individuals in each data set a unique person identifier, called a Protected Identification Key (PIK). CARRA assigned these unique identifiers via the Person Identification Validation System (PVS), which employs probabilistic record linkage techniques (see Wagner and Layne, 2014, for more information). CARRA uses personally identifiable information (PII) such as Social Security Number, name, date of birth, and address to assign a PIK by comparing the same fields in a master reference file constructed from federal administrative data sources. CARRA then removes the PII from the data file to anonymize the data and preserve confidentiality so it can be used for statistical purposes and research. Only those observations that received the unique key are used in the analysis.

The CPS provides the identification of tipped workers in the restaurant industry through the 4-digit occupation code, as well as the demographic, labor, and state characteristics used in the econometric analysis. I retain all of the respondents who report being a full-service restaurant server: wait staff and bartenders. Full wages for these individuals come from the Wages, Salary, and Tips field of the W-2. I also retain the variable that reflects FICA wage—this is wage paid by employers directly to the employee. Subtracting the latter variable from total wages, salary, and tips yields tip income.

A key concern is how hourly wage is calculated: I take yearly W-2 earnings and divide them by hours worked per year. Hours worked per year is itself the product of weeks worked per year and usual hours worked per week. Because wage floors may impact both hourly wage and the supply of labor over the year, this calculation is necessary. However, even assuming that the W-2 FICA wage accurately records wages from employers over the year, some error may be introduced in the hourly wage equivalent if weeks worked per year or usual hours per week are recorded with error.

Another concern is the number of employers a person may have during the year. A W-2 should exist for each job held. If more than one job is held simultaneously, and one of those jobs is a non-serving job, then FICA wages will reflect both tipped and nontipped wages. To avoid this, I take only earners who held one job at a time, and took only jobs in which FICA wages were less than Wages, Salary, and Tips. Those who held

only one job at a time were those whose number of W-2s matched the number of employers they reporting having, non-simultaneously, over the year. This retains approximately half of those whose specified occupation is restaurant server. <sup>9</sup>

Appendix table 1 provides a full overview of year-by-state minimum wages, showing the average FICA wage and total wage in each state. For the most part, the listed average wage for each state and year looks reasonable, with average FICA wages generally above the tipped minimum wage, and a higher FICA wage in states with higher minimum wages. In many cases, the average FICA wage is close in value to the minimum wage, indicating that when the wage floor increases, the increase will "bite" at the low end of the wage distribution.

#### 3.3 An assessment of W-2 and survey hourly wage

To assess how survey reports and W-2 administrative records compare, I linked CPS ASEC participants to their outgoing rotation group data for the same year.<sup>10</sup> When respondents rotate out of the survey sample, they receive questions on their current employment situation. Questions include: whether the respondent works by the hour; how much he or she makes per hour in non-tipped wages; whether he or she receives tips; and how much in total compensation he or she makes per week. The measurement of hourly wage from the linked CPS ASEC/ORG–W-2 data is based on the W-2 annual wage information divided by hours worked per year. Meanwhile, the hourly rate in the outgoing rotation group is a direct measure taken in the month that the respondent leaves the survey. Respondents are prompted to report what they make per hour without including tips or commissions.

Figure 2 shows kernel density estimates of ORG hourly wage as a function of W-

<sup>&</sup>lt;sup>9</sup>In a sensitivity test, I lifted the second restriction, keeping everyone whose number of W-2s matched their number of employers. Results were slightly weaker, but qualitatively unchanged.

<sup>&</sup>lt;sup>10</sup>Further work on the measurement of W-2 information will use both the same-year outgoing responses and responses for the preceding rotation when it is available. See http://www.census.gov/prod/2006pubs/tp-66.pdf for the rotation structure of the CPS.

2 constructed hourly wage (top panel) and ORG earnings as a function of W-2 FICA wages divided by weeks worked. The results of this simple analysis are striking, with hourly earnings derived from the survey suffering from marked over reporting below approximately \$7 or \$8 per hour, and suffering from underreporting beyond that point. Estimates get very noisy beyond about \$14 or \$15 per hour. Weekly earnings, shown in the bottom panel, are slightly less prone to error.

There are several reasons why the two sources of information may not match. The most obvious reason is that wages, earnings, and income are routinely reported with error by respondents—a subject of multiple investigations (see ?, for example). If we take W-2 wages as "true," the error shown in these graphs follows the general pattern, but is an extreme example, of wage misreporting—overestimation at the low end and underestimation at the high end. Second, as few as five and as many as 13 months passed between the generation of the information that makes up the W-2 measure and the ORG information. A respondent may have changed jobs or occupations during this time, may have received a pay increase, or may have left the labor force entirely. Finally, because weekly earnings appear to be reported more accurately than non-tipped hourly wages, it is possible that FICA wages are not as "clean" as they should be, but that some employers are including tip income in their FICA reporting. A conclusion to draw from this analysis is that W-2 reporting should be more thoroughly investigated, which will be the focus of a separate paper.

## 4 Theoretical and empirical model

### 4.1 Theory

Wessels examined the market for tipped servers first in a competitive model (1993) and then as a monopsony (1997). Wessels' argument that the hiring of tipped servers is an example of monopsony power is a compelling one: for a given number of tables, hiring an additional server decreases per-person tips for that shift. Thus, a restaurant owner must offer more pay to all servers when hiring an additional server. Because the marginal cost of servers is thus greater than the average cost, the market fits the classic definition of a monopsony.

Figure 3 shows Wessels' model for the restaurant industry. On the *x* axis is serving time per meal, which will reflect the number of servers for a set quality and quantity of meals and constant returns to scale. As *S* increases, the number of meals served in an hour decreases and the hourly tips decrease. On the *y* axis is the wage paid to a server per hour. Restaurants face a server wage of  $W_s$  but can manipulate payment versus service by having more or fewer servers working on each shift. Thus, the restaurant actually faces a supply for "serving time per meal" of *W*, and servers are paid W + tP/S, where tP/S expresses the tips per hour as a function of the price of the meal and the number served. The customers' "demand price of service time" is shown as curve  $P(1 + t)MP_s$ .

In the absence of a wage floor, a restaurant owner will hire servers to fulfill a service per meal at A, which corresponds to the wage  $w_m$  (the intersection of the wage plus tips and the demand for service curve). Were there to be a wage imposed between  $w_m$  and  $w_c$  (the wage in a competitive market), hiring and service per meal would increase. Lower tips would offset the wage increase, and servers would not be better off monetarily (although they might benefit from better table coverage). A tipped minimum wage above  $w_c$  causes hiring to decrease back along the service demand curve.

In a competitive model, the imposition of a wage floor is predicted to decrease employment or increase output prices, or both. In a monopsony where the wage floor is set between the monopsony wage and the competitive wage, we would expect to see employment increase and output prices decrease. However, the situation becomes muddy when the portion of the wage paid as tips is a function of the output price. Wessels recognized this element of tipped wages in his model, but assumed a constant per-worker

effort. However, restaurants are defined by their seating capacity and hours of operation. For a given number of tables, a restauranteur may curtail serving hours or increase the number of tables per server per shift rather than cut employees or raise prices. The existence of tip pooling and service charges add to the manner in which employers may react to higher tipped minimums.

That being said, the key testable hypothesis from this model is: tipped minimum wages should increase the portion of a server's per-hour wage that comes from the employer. If wages are already higher than the a proposed tipped minimum for all servers, then an increase in the statutory wage will have no effect. With W-2 data, this hypothesis should be easy to test, as the FICA wage field is the portion of a server's wages paid directly from the employer. The second hypothesis is that a restaurant will hire more servers, and per-person tips will decrease. Wessels uses restaurant employment divided by restaurant sales, while acknowledging that a better measure would be total person hours of servers. Even better is a direct measure of hourly tips, which is also available from the W-2. Finally, the monopsony model predicts that employment should increase until point  $w_c$  on the graph, and then decrease. This can be tested using total server employment divided by the population of workers who identify as servers in the CPS ASEC.

### 4.2 Empirics

To examine the evidence on tipped minimum wages using the theoretical construct outlined above, I use a difference-in-differences approach that has become standard in analyses of this type. First, I collapse the data into state-year cells, retaining information on tipped minimum wages, minimum wages, and rates of employment and demographic characteristics. An effect of higher minimum wages is identified using the variation in state tipped minimums over time. The dependent variables in question (FICA wages, total compensation including tips, employment, and hours worked) is examined in turn using the following model:

$$\ln(y_{st}) = \alpha_{st} + \beta * \ln(TW_{st}) + \gamma * \ln(MW_{st}) + \phi_s + \eta_t + \epsilon_{st}$$
(1)

where  $\ln(y_{st})$  is one of the dependent variables described above,  $\ln(TW_{st})$  and  $\ln(MW_{st})$  are the tipped minimum wage and minimum wage in state *s* and time *t*, and  $\phi_s$  and  $\eta_t$  are state and time fixed effects. To this baseline model, I then add demographic characteristics measured at the state level, including log total employment, average wage for all workers, and the proportion of the labor force in the state who are men, the proportion married, the proportion with a high school degree or more, and the proportion of nonwhites. I add information on restaurant prices by including the "food away from home" component of the Bureau of Labor Statistics' Consumer Price Index. The last model includes a state-specific time trend, with the caveat that the model may not fit as well at others due to near collinearity.<sup>11</sup> All specifications employ standard errors clustered at the state level, and each regression is weighted using the number of observations in the state-year cell.

### 5 Results

Table 1 provides summary statistics for the sample used in the regression analysis. Overall, the mean tipped minimum wage is \$3.19 per hour, compared with a mean FICA wage of \$4.44 per hour. The mean minimum wage is \$6.43 per hour, while the per hour total compensation for servers is \$7.42. These numbers give some indication that, on average at least, state tip credits and actual compensation received are in line with states' minimum wages. The low pay of servers is contrasted with the average hourly wage for all workers of \$21.79 per hour.

<sup>&</sup>lt;sup>11</sup>The number of year-state observations limited to restaurant servers leads to an analysis data set of fewer than 400 observations.

The average size of the server labor force within a state cell is approximately 14. This may present some problems for the analysis if results are driven by small cell sizes. I experimented with different limitations on the analysis. Results did not change much when the cell size was limited to greater than 3 compared with greater than 5. The ratio of servers to non-servers in the restaurant industry is approximately 0.3. I did not include restaurant employees who may be partially tipped in the server category, including food runners, hosts and hostesses, and busers; according to the IRS-provided definition, these employees should be covered by the minimum wage and not the tipped minimum.

Table 2 shows the first set of results of the difference-in-differences specifications. I find strong evidence that hourly wages paid by restaurant owners to their servers increase with increases in the tipped minimum wage. An elasticity of approximately 0.5 to 0.6 indicates that when the tipped minimum increases by 10 percent, FICA wages of servers increase by 5 to 6 percent, meaning that tipped minimum increases affect a significant proportion of tipped workers. These results are in line with the evidence presented in Appendix table 1, in that many of the states with higher tipped minimum wages do appear to have a higher mean FICA wage. These results are also in line with the wide variation in tipped minimums across the states. Meanwhile, the state minimum wage does not appear to affect the wages of tipped employees, which is as expected.

In contrast, I do not find that overall compensation increases with increases in the tipped minimum wage—this is due to tips decreasing in response to tipped minimum wage increases. Tips per hour appear to decrease in response to higher tipped minimum wages in the same proportion as FICA wage increase (5 or 6 percent). Taken alone, these results are consistent with the monopsony model, as we expect less in tips as employer-paid wage increases. However, we should also expect employment to increase at the same time, which would account for lower tips resulting from greater table coverage.

Turning to table 3, we see that in a linear model there is no relationship between tipped minimum wages and server employment. This is true for both person employment and hours worked contingent on employment. However, with such a wide range of tipped minimums, the possibility exists that employment first rises and then falls as the tipped minimum wage is increased. Looking at a partial residual plot of the full model, it appears that the relationship between the tipped minimum wage (logged) and the rate of server employment is certainly not linear, and might be quadratic. While the graph looks noisy, the resulting quadratic estimation is presented in Table 4. The coefficients on the tipped minimum wage and its square are statistically significant in the first two models, with the interpretation that the relationship is positive over lower values of the tipped minimum wage but eventually level out and then decrease. When controlling for state-specific time trends, the coefficients are smaller and no longer statistically significant. There is no relationship between hours worked and tipped minimum wages in the quadratic equation.

The coefficients from the quadratic model imply that as tipped minimum wages increase from \$2.13 to about \$4.50, employment in the server sector increases.<sup>12</sup> Yet, considering how strong the elasticities are for both wages and tips, the employment evidence is noisier and not as strong. It is more than likely that some employers respond to higher wage floors in other ways. In practice, one explanation for lower tips could be that employers institute or intensify tip-pooling schemes to apply mandated wage increases more equitably across their employees. When an employer must raise wages for servers, he or she may feel it necessary to compensate partially tipped employees who are not officially covered by the mandate in order to retain workers who might otherwise consider the wage increase unfair. There is also some anecdotal evidence that employers may turn to service charges when tipped minimum wages increase (Azar, 2012). If these schemes are more common in high tipped-minimum-wage states, it may

<sup>&</sup>lt;sup>12</sup>In 2011, 13 states had tipped minimum wages higher than \$4.50 per hour.

put downward pressure on tip income. For example, an employer may begin to apply a service charge to all tables over a certain number. These service charges are owned by the employer and can be used to meet tip credit requirements, regardless of which employee actually served the table. This may be especially problematic if employers then report these tips in the FICA wage field, since it will then represent a simple shift of wages from tips to FICA wage. In all cases, the combined results make it appear that servers do not see an increase in overall compensation when a tipped minimum wage is established, but do experience greater employment over a substantial proportion of state minimum wages.

As a sensitivity check, it has become a common practice in minimum wage papers to randomly assign wage floors to states in the data and see whether statistically significant results can be estimated from these false wage floors. Following this procedure, I ran each specification presented in the main results in which a statistically significant association was found, using a randomly assigned placebo tipped minimum wage. There is no statistically significant relationship between the placebo and any dependent variable used in the analysis.

### 6 Conclusion

This paper presents a new source of information for wages on tipped servers and analyzed the effect of the tipped minimum wage on wages, hourly tips, server employment, and hours worked. It also provides some assessment of the quality of the data and compared it with similar measures from the CPS ORG. I present evidence that raising the tipped minimum wage has the effect of raising that portion of wages paid by employers, but decreases tip income by a similar percentage. An increase in the minimum wage has a quadratic relationship with sector employment, where employment first increases, levels out, and then decreases as the tipped minimum increases. This result is consistent with a monopsony model, with some caveats about the noisiness of the employment estimates compared with the wage and tip estimates.

A limitation of this work is that, while we can feel confident that W-2 FICA wages represent what a person was paid directly by his or her employer, CPS-reported hours per year must be used to calculate an hourly equivalent wage. If hours and weeks of work for the year suffer from error, the calculated hourly wage will also suffer. A second limitation is that the results may be driven by employers in higher tipped-minimumwage state moving to tip-pooling or service-charge schemes. If these tips are reported in the FICA wage part of the W-2, the effect of the tipped minimum wage will be overstated.

That being said, the paper makes a contribution to the literature on minimum wages by using administrative records to generate separate measures of employer-paid wages and tips-measures that have proven difficult to capture in other sources of data. The clear demonstration of a monopsony effect provides some policy guidance on the setting of minimum wages, with the understanding that higher minimums may induce employment in industries where a monopsony may exist. Finally, the results provide evidence on the welfare effects of tipped minimum wages on servers, showing that, while overall compensation may not change, servers benefit from higher employment and thus better table coverage over a wide range of the tipped minimum wage.

More work needs to be done on assessing the wage information generated by the W-2. An especially useful activity would involve discovery of actual reporting practices (that is, how well employers follow the W-2 instructions), how tip-pooling and service charges affect reporting, and how these activities are affected by minimum wage increases.

## References

- Sylvia A Allegretto. Waiting for change: Is it time to increase the \$2.13 subminimum wage? 2013.
- John E Anderson and Örn B Bodvarsson. Do higher tipped minimum wages boost server pay? *Applied Economics Letters*, 12(7):391–393, 2005.
- Ofer H Azar. The effect of the minimum wage for tipped workers on firm strategy, employees and social welfare. *Labour Economics*, 19(5):748–755, 2012.
- William M. Boal and Michael R. Ransom. Monopsony in the labor market. *Journal of Economic Literature*, 35(1), 1997.
- Charles Brown. Minimum wages, employment, and the distribution of income. *Handbook of labor economics*, 3:2101–2163, 1999.
- Arindrajit Dube, T William Lester, and Michael Reich. Minimum wage effects across state borders: Estimates using contiguous counties. *The review of economics and statistics*, 92 (4):945–964, 2010.
- William E Even and David A Macpherson. The effect of the tipped minimum wage on employees in the us restaurant industry. *Southern Economic Journal*, 80(3):633–655, 2014.
- Jeffrey C Moore, Linda L Stinson, and Edward J Welniak. Income measurement error in surveys: A review. *JOURNAL OF OFFICIAL STATISTICS-STOCKHOLM-*, 16(4):331–362, 2000.
- Multiple. Behind the kitchen door: A multi-site study of the restaurant industry. Technical report, ROCUnited, 2011.
- David Neumark and William Wascher. Minimum wages and employment: A review of evidence from the new minimum wage research. Technical report, National Bureau of Economic Research, 2006.
- David Neumark, JM Salas, and William Wascher. Revisiting the minimum wageemployment debate: Throwing out the baby with the bathwater? Technical report, National Bureau of Economic Research, 2013.
- Harold S Peckron. Tip police: Aftermath of the fior d'italia rule, the. *Cath. UL Rev.*, 52:1, 2002.
- Terrence Bouvier Robinson. A comprehensive look at the tip wage. *Compensation & Benefits Review*, 43(4):214–219, 2011.
- Willard L Rodgers, Charles Brown, and Greg J Duncan. Errors in survey reports of earnings, hours worked, and hourly wages. *Journal of the American Statistical Association*, 88 (424):1208–1218, 1993.

- Deb Wagner and Mary Layne. The Person Identification Validation System (PVS). Technical report, U.S. Census Bureau, Center for Administrative Records Research and Applications, 2014.
- Walter John Wessels. The minimum wage and tipped employees. *Journal of Labor Research*, 14(3):213–226, 1993.
- Walter John Wessels. Minimum wages and tipped servers. *Economic Inquiry*, 35(2):334–349, 1997.
- Siobhann Kathleen Williams. Tips on tip reporting: Are you at risk for back fica taxes?, 2014. URL http://www.restaurantowner.com/public/342.cfm.



Figure 1: The maps show states with tipped minimum wage for 2005 (top) and 2011 (bottom). Graph shading indicates the federal minimum wage times 1 (white) to more than 4 times (darkest blue).



Figure 2: The top figure shows a kernel density estimates of ORG hourly wage (not including tips) as a function of FICA wage calculated from W-2 total FICA earnings for the year divided by hours worked per year. The bottom figure is an estimate of ORG weekly compensation (including tips) as a function of W-2 Wage, Salary, and Tips divided by weeks worked per year. See text for description of variables.



Figure 3: Adapted from Wessels (1997). The figure shows the monopsony market for tipped servers. See text for description.

]	lat	ole	1:	Summary	statistic	cs of	varial	oles	used	in	anal	ysis	3

State characteristics	Mean	SD
Tipped minimum wage	3.19	1.55
Minimum wage	6.43	1.17
Employment rate	0.94	0.02
Size of labor force, all workers	586.93	1.64
Price of meals	8.28	1.13
Average hourly wage, all workers	21.79	1.31
Proportion male	0.57	0.03
Proportion married	0.63	0.05
Proportion nonwhite	0.30	0.18
Proportion with HS education +	0.75	0.14
State characteristics, servers	$\land$ >	
FICA wage per hour	4.44	0.32
Size of server labor force	13.61	1.81
Annual hours	871.47	1.32
Total hourly compensation	7.42	0.29
Ratio of servers to non-server restaurant employment	0.25	0.07
Observations	357	

Source: Linked CPS ASEC–W-2 file for tax years 2005 to 2011. The columns show the mean and standard deviation for variables after collapsing by state and year. The number of observations reflects the number of year-state cells (51 states including DC and 7 years). Key variables from the W-2 are FICA wage per hour and total compensation per hour, which are arrived at by diving the separate values reported on the W-2 by total hours worked in the year, reported in the CPS ASEC.

				0	1	
	FI	CA Wag	e	Tip	os per H	our
Tipped minimum wage (log)	0.57***	0.59***	0.54*	-0.55*	-0.56**	-0.82*
	(0.14)	(0.13)	(0.21)	(0.21)	(0.18)	(0.37)
Minimum wage (log)	0.10	-0.01	-0.07	-0.06	-0.36	-0.36
0 . 0,	(0.20)	(0.14)	(0.23)	(0.31)	(0.29)	(0.40)
Employment rate, all worker	· /	0.40	-0.66	. ,	0.91	1.40
1		(0.97)	(1.38)		(1.56)	(3.22)
Population (log)		0.82**	0.61		-0.69	-0.50
1 (0)		(0.29)	(0.41)		(0.64)	(0.95)
Price of meals (log)		-0.28	-0.39		0.33	0.49
		(0.19)	(0.29)	JX	(0.33)	(0.46)
Average hourly wage, all workers (log)		0.85***	0.89***	*	1.18**	* 1.20***
		(0.05)	(0.06)	$\sim$	(0.16)	(0.20)
Proportion male		-0.08	-0.02	$\mathbf{)}$	-0.29	-0.55
*	~	(0.58)	(0.74)		(1.28)	(1.82)
Proportion married		0.04	0.25	<b>X</b>	-0.03	-0.26
-		(0.54)	(0.71)		(1.47)	(1.93)
Proportion nonwhite		0.31	0.42		0.05	-0.07
		(0.42)	(0.52)		(0.88)	(1.26)
Proportion with HS education+	$\mathcal{I}$	-0.25*	-0.15		$0.44^{*}$	0.27
		(0.10)	(0.13)		(0.20)	(0.27)
Constant	0.13	-9.53**	-6.70	1.35*	5.36	3.87
	(0.36)	(3.10)	(4.57)	(0.52)	(6.54)	(10.10)
State dummies	Yes	Yes	Yes	Yes	Yes	Yes
Year dummies	Yes	Yes	Yes	Yes	Yes	Yes
Trends	No	No	Yes	No	No	Yes
Observations		357			357	
R2	0.67	0.68	0.72	0.56	0.57	0.61

Table 2: Difference-in-difference estimates: wages and tips

Source: Linked CPS ASEC–W-2 file for tax years 2005 to 2011.

p < 0.05, p < 0.01, p < 0.01, p < 0.001. The table shows the results of difference-in-differences specifications for FICA wage, full hourly compensations, and employment rate of servers. Model 1 includes only the tipped minimum wage and the minimum wage in the specification, where the minimums are the higher of the federal or state minimum, and state and year fixed effects. Model 2 shows the same specification, but includes covariates measured at the state level. Model 3 includes state-specific time trends. All models use standard errors clustered at the state level.

	Sever employment Hours worked		worked r	per vear		
						, cui
Tipped minimum wage (log)	0.07	0.06	0.08	-0.27	-0.24	-0.44
	(0.06)	(0.06)	(0.07)	(0.16)	(0.15)	(0.31)
Minimum wage (log)	-0.03	-0.02	-0.07	0.29	0.30	0.47
	(0.06)	(0.06)	(0.06)	(0.27)	(0.24)	(0.34)
Employment rate,		0.07	-0.23		-0.96	-1.47
		(0.41)	(0.53)		(1.11)	(1.68)
Population (log)		-0.12	0.01		-0.97*	-1.41*
		(0.16)	(0.17)		(0.43)	(0.61)
Price of meals (log)		-0.05	-0.05		0.05	0.34
		(0.10)	(0.13)		(0.29)	(0.39)
Average hourly wage, all workers (log)		0.00	0.00		-0.37***	-0.35***
		(0.02)	(0.02)	$\sim$	(0.07)	(0.08)
Proportion male		-0.02	-0.04	JY	0.92	1.81
		(0.28)	(0.34)		(0.77)	(1.03)
Proportion married		0.18	0.19	<b>X</b>	-0.18	-0.34
-		(0.25)	(0.30)		(0.82)	(0.99)
Proportion nonwhite		-0.03	0.00		-0.09	0.45
		(0.16)	(0.26)		(0.50)	(0.52)
Proportion with HS education or more	$\mathcal{I}$	-0.00	-0.05		0.60***	0.53***
		(0.04)	(0.04)		(0.12)	(0.15)
Constant	0.25*	1.23	0.31	6.60**	* 17.11***	19.93**
	(0.10)	(1.69)	(1.69)	(0.47)	(4.63)	(6.46)
State dummies	Yes	Yes	Yes	Yes	Yes	Yes
Year dummies	Yes	Yes	Yes	Yes	Yes	Yes
Trends	No	No	Yes	No	No	Yes
Observations		357			357	
R2	0.67	0.68	0.72	0.56	0.57	0.61

Table 3: Difference-in-difference estimates: employment effects

Source: Linked CPS ASEC–W-2 file for tax years 2005 to 2011.

p < 0.05, p < 0.01, p < 0.01, p < 0.001. The table shows the results of difference-in-differences specifications for FICA wage, full hourly compensations, and employment rate of servers. Model 1 includes only the tipped minimum wage and the minimum wage in the specification, where the minimums are the higher of the federal or state minimum, and state and year fixed effects. Model 2 shows the same specification, but includes covariates measured at the state level. Model 3 includes state-specific time trends. All models use standard errors clustered at the state level.

	Seve	employ	ment
Tipped minimum wage (log)	0.35*	0.37*	0.09
	(0.15)	(0.18)	(0.30)
Minimum wage (log)	-0.04	-0.04	-0.07
	(0.06)	(0.06)	(0.06)
Square of tipped minimum (log)	-0.12*	-0.13*	-0.01
	(0.05)	(0.06)	(0.13)
Employment rate, all workers		-0.27	-0.23
		(0.38)	(0.52)
Population (log)		-0.13	0.01
		(0.15)	(0.17)
Price of meals (log)		-0.03	-0.05
		(0.09)	(0.13)
Average hourly wage, all workers (log)		0.00	0.00
		(0.02)	(0.02)
Proportion male		-0.07	-0.04
		(0.28)	(0.34)
Proportion married		0.17	0.19
		(0.25)	(0.30)
Proportion nonwhite		0.02	0.00
	$\mathcal{I}$	(0.18)	(0.26)
Proportion with HS education or more		-0.02	-0.05
		(0.04)	(0.05)
Constant	0.12	1.58	0.31
	(0.11)	(1.60)	(1.70)
State dummies	Yes	Yes	Yes
Year dummies	Yes	Yes	Yes
Trends	No	No	Yes
Observations		357	
R2	0.67	0.68	0.72

Table 4: Difference-in-difference estimates: Rate of employment of waitstaff, alternate specification

Source: Linked CPS ASEC–W-2 file for tax years 2005 to 2011. \*p < 0.05, \*\*p < 0.01, \*\*\*p < 0.001. The table shows the results of difference-in-differences specifications for employment rate of servers using a quadratic in the log tipped minimum wage. Model 1 includes only the tipped minimum wage and the minimum wage in the specification, where the minimums are the higher of the federal or state minimum, and state and year fixed effects. Model 2 shows the same specification, but includes covariates measured at the state level. Model 3 includes state-specific time trends. All models use standard errors clustered at the state level.



Figure 4: The graphs show partial residual plots of the relationship between the tipped minimum wage and employment rate among servers (top) and graphs out quadratic relationship over all values of the logged tipped minimum wage (bottom).

### 7 Appendix

The following pages show the IRS instructions for employers in filling out employees' W-2s. Information pertaining to tips is highlighted.

Form W-2c showing the employee's SSN. If the employee needs to change his or her name from that shown on the card, the employee should call the SSA at 1-800-772-1213.

If you do not provide the correct employee name and SSN on Form W-2, you may owe a penalty unless you have reasonable cause. For more information, see Publication 1586, Reasonable Cause Regulations & Requirements for Missing and Incorrect Name/TINs.

ITINs for aliens. Do not accept an ITIN in place of an SSN for employee identification or for work. An ITIN is only available to resident and nonresident aliens who are not eligible for U.S. employment and need identification for other tax purposes. You can identify an ITIN because it is a 9-digit number, beginning with the number "9" with either a "7" or "8" as the fourth digit, and is formatted like an SSN (for example, 9NN-7N-NNNN). An individual with an ITIN who later becomes eligible to work in the United States must obtain an SSN.

Do not auto-populate an ITIN into box a.



Box b—Employer identification number (EIN). Show the EIN assigned to you by the IRS (00-0000000). This should be the same number that you used on your federal employment tax returns (Forms 941, 941-SS, 943, 944, CT-1, or Schedule H (Form 1040)). Do not use a prior owner's EIN. If you do not have an EIN when filing Forms W-2, enter "Applied For" in box b; do not use your SSN. You can get an EIN by applying online at IRS.gov, by calling the toll-free number, 1-800-829-4933, or by filing Form SS-4, Application for Employer Identification Number. Also see Agent reporting.

Box c—Employer's name, address, and ZIP code. This entry should be the same as shown on your Forms 941, 941-SS, 943, 944, CT-1, or Schedule H (Form 1040). The U.S. Postal Service recommends that no commas or periods be used in return addresses. Also see Agent reporting.

Box d—Control number. You may use this box to identify individual Forms W-2. You do not have to use this box.

Boxes e and f—Employee's name and address. Enter the name as shown on your employee's social security card (first name, middle initial, last name). If the name does not fit in the space allowed on the form, you may show the first and middle name initials and the full last name. It is especially important to report the exact last name of the employee. If you are unable to determine the correct last name, use of the SSA's Social Security Number Verification System may be helpful. Separate parts of a compound name with either a hyphen or a blank. Do not join them into a single word. Include all parts of a compound name in the appropriate name field. For example, for the name "John R Smith-Jones," enter "Smith-Jones" or "Smith Jones" in the last name field. If the name has changed, the employee must get a corrected social security card from any SSA office. Use the name on the original card until you see the corrected card. Do not show titles or academic degrees, such as "Dr.," "RN," or "Esq.," at the beginning or end of the employee's name. Generally, do not enter "Jr.," "Sr.," or other suffix in the "Suff." box on Copy A unless the suffix appears on the card. However, the SSA still prefers that you do not enter the suffix on Copy A.

Include in the address the number, street, and apartment or suite number (or P.O. box number if mail is not delivered to a street address). The U.S. Postal Service recommends that no commas or periods be used in delivery addresses. For a foreign address, give the information in the following order: city, province or state, and country. Follow the country's practice for entering the postal code. Do not abbreviate the country name.

The IRS expects to change the third-party sick pay recap reporting and filing requirements for wages paid in 2014. Information about this change will be included in the 2014 Publication 15-A, Employer's Supplemental Tax Guide (Supplement to Circular E, Employer's Tax Guide, Publication 15) and other relevant tax products.

Box 1—Wages, tips, other compensation. Show the total taxable wages, tips, and other compensation (before any payroll deductions) that you paid to your employee during the year. However, do not include elective deferrals (such as employee contributions to a section 401(k) or 403(b) plan) except section 501(c)(18) contributions. Include the following.

 Total wages, bonuses (including signing bonuses), prizes, and awards paid to employees during the year.
 See Calendar year basis.

2. Total noncash payments, including certain fringe benefits. See *Fringe benefits*.

3. Total tips reported by the employee to the employer (not allocated tips).

4. Certain employee business expense

reimbursements. See Employee business expense reimbursements.

5. The cost of accident and health insurance premiums for 2%-or-more shareholder-employees paid by an S corporation.

6. Taxable benefits from a section 125 (cafeteria) plan if the employee chooses cash.

7. Employee contributions to an Archer MSA.

8. Employer contributions to an Archer MSA if includible in the income of the employee. See Archer

MSA.

 Employer contributions for qualified long-term care services to the extent that such coverage is provided through a flexible spending or similar arrangement.

10. Taxable cost of group-term life insurance in excess of \$50,000. See *Group-term life insurance.* 

11. Unless excludable under *Educational assistance* programs, payments for non-job-related education expenses or for payments under a nonaccountable plan. See Pub. 970.

12. The amount includible as wages because you paid your employee's share of social security and Medicare taxes (or railroad retirement taxes, if applicable). See *Employee's social security and Medicare taxes (or railroad retirement taxes, if applicable) paid by employer.* 

-14- General Instructions for Forms W-2 and W-3 (2014)

If you also paid your employee's income tax withholding, treat the grossed-up amount of that withholding as supplemental wages and report those wages in boxes 1, 3, 5, and 7. (Use box 14 if railroad retirement taxes apply.) No exceptions to this treatment apply to household or agricultural wages.

13. Designated Roth contributions made under a section 401(k) plan, a section 403(b) salary reduction agreement, or a governmental section 457(b) plan. See Designated Roth contributions.

14. Distributions to an employee or former employee from an NQDC plan (including a rabbi trust) or a nongovernmental section 457(b) plan.

15. Amounts includible in income under section 457(f) because the amounts are no longer subject to a substantial risk of forfeiture.

16. Payments to statutory employees who are subject to social security and Medicare taxes but not subject to federal income tax withholding must be shown in box 1 as other compensation. See *Statutory employee*.

17. Cost of current insurance protection under a

compensatory split-dollar life insurance arrangement. 18. Employee contributions to a health savings account (HSA).

19. Employer contributions to an HSA if includible in the income of the employee. See *Health savings account* (HSA).

20. Amounts includible in income under an NQDC plan because of section 409A. See *Nonqualified deferred compensation plans.* 

21. Payments made to former employees while they are on active duty in the Armed Forces or other uniformed services.

22. All other compensation, including certain scholarship and fellowship grants. See *Scholarship and fellowship grants*. Other compensation includes taxable amounts that you paid to your employee from which federal income tax was not withheld. You may show other compensation on a separate Form W-2. See *Multiple forms*.

Box 2—Federal income tax withheld. Show the total federal income tax withheld from the employee's wages for the year. Include the 20% excise tax withheld on excess parachute payments. See *Golden parachute payments*.

For Forms W-2AS, W-2CM, W-2GU, or W-2VI, show the total American Samoa, CNMI, Guam, or U.S. Virgin Islands income tax withheld.

Box 3—Social security wages. Show the total wages paid (before payroll deductions) subject to employee social security tax but not including social security tips and allocated tips. If reporting these amounts in a subsequent year (due to lapse of risk of forfeiture), the amount must be adjusted by any gain or loss. See Box 7—Social security tips and Box 8—Allocated tips. Generally, noncash payments are considered to be wages. Include employee business expense reimbursements reported in box 1. If you paid the employee's share of social security and Medicare taxes rather than deducting them from

General Instructions for Forms W-2 and W-3 (2014)

wages, see Employee's social security and Medicare taxes (or railroad retirement taxes, if applicable) paid by employer. The total of boxes 3 and 7 cannot exceed \$117,000 (2014 maximum social security wage base).

Report in box 3 elective deferrals to certain qualified cash or deferred compensation arrangements and to retirement plans described in box 12 (codes D, E, F, G, and S) even though the deferrals are not includible in box 1. Also report in box 3 designated Roth contributions made under a section 401(k) plan, under a section 403(b) salary reduction agreement, or under a governmental section 457(b) plan described in box 12 (codes AA, BB, and EE).

Amounts deferred (plus earnings or less losses) under a section 457(f) or nonqualified plan or nongovernmental section 457(b) plan must be included in boxes 3 and/or 5 as social security and/or Medicare wages as of the later of when the services giving rise to the deferral are performed or when there is no substantial forfeiture risk of the rights to the deferred amount. Include both elective and nonelective deferrals for purposes of nongovernmental section 457(b) plans.



 Signing bonuses an employer pays for signing or ratifying an employment contract. See Rev. Rul. 2004-109, 2004-50 I.R.B 958, available at <u>www.irs.gov/</u> irb/2004-50 IRB/ar07.html.

Taxable cost of group-term life insurance over \$50,000 included in box 1. See Group-term life insurance.
Cost of accident and health insurance premiums for 2%-or-more shareholder-employees paid by an S corporation, but only if not excludable under section 3121(a)(2)(B).

 Employee and nonexcludable employer contributions to an MSA or HSA. However, do not include employee contributions to an HSA that were made through a cafeteria plan. See Archer MSA and Health savings account (HSA).

• Employee contributions to a SIMPLE retirement account. See SIMPLE retirement account.

Adoption benefits. See Adoption benefits.

Box 4—Social security tax withheld. Show the total employee social security tax (not your share) withheld, including social security tax on tips. For 2014, the amount should not exceed \$7,254 (\$117,000 × 6.2%). Include only taxes withheld (or paid by you for the employee) for 2014 wages and tips. If you paid your employee's share, see Employee's social security and Medicare taxes (or railroad retirement taxes, if applicable) paid by employer.

Box 5—Medicare wages and tips. The wages and tips subject to Medicare tax are the same as those subject to social security tax (boxes 3 and 7) except that there is no wage base limit for Medicare tax. Enter the total Medicare wages and tips in box 5. Be sure to enter tips that the employee reported even if you did not have enough employee funds to collect the Medicare tax for those tips. See Box 3—Social security wages, for payments to report in this box. If you paid your employee's share of taxes, see

-15-

Employee's social security and Medicare taxes (or railroad retirement taxes, if applicable) paid by employer.

If you are a federal, state, or local governmental agency with employees paying only Medicare tax, enter the Medicare wages in this box. See Government employers.

Example of how to report social security and Medicare wages. You paid your employee \$140,000 in wages. Enter in box 3 (social security wages) 117000 but enter in box 5 (Medicare wages and tips) 140000.00. There is no limit on the amount reported in box 5. If the amount of wages paid was \$117,000 or less, the amounts entered in boxes 3 and 5 would be the same

Box 6-Medicare tax withheld. Enter the total employee Medicare tax (including any Additional Medicare Tax) withheld. Do not include your share Include only tax withheld for 2014 wages and tips. If you paid your employee's share of the taxes, see Employee's social security and Medicare taxes (or railroad retirement taxes, if applicable) paid by employer.

For more information on Additional Medicare Tax, go to IRS.gov and enter "Additional Medicare Tax" in the search box.

Box 7-Social security tips. Show the tips that the employee reported to you even if you did not have enough employee funds to collect the social security tax for the tips. The total of boxes 3 and 7 should not be more than \$117,000 (the maximum social security wage base for 2014). Report all tips in box 1 along with wages and other compensation. Include any tips reported in box 7 in box 5 also

Box 8—Allocated tips (not applicable to Forms W-2AS, W-2CM, W-2GU, or W-2VI). If you are a food or beverage establishment, show the tips allocated to the employee. See the Instructions for Form 8027, Employer's Annual Information Return of Tip Income and Allocated Tips. Do not include this amount in boxes 1, 3, 5, or 7.

Box 9. Do not enter an amount in box 9

Box 10—Dependent care benefits (not applicable to Forms W-2AS, W-2CM, W-2GU, or W-2VI). Show the total dependent care benefits under a dependent care assistance program (section 129) paid or incurred by you for your employee. Include the fair market value (FMV) of care in a daycare facility provided or sponsored by you for your employee and amounts paid or incurred for dependent care assistance in a section 125 (cafeteria) plan. Report all amounts paid or incurred (regardless of any employee forfeitures), including those in excess of the \$5,000 exclusion. This may include (a) the FMV of benefits provided in kind by the employer, (b) an amount paid directly to a daycare facility by the employer or reimbursed to the employee to subsidize the benefit, or (c) reimoursed to the employee to substatize the benefit, or benefits from the pre-tax contributions made by the employee under a section 125 dependent care flexible spending account. Include any amounts over \$5,000 in boxes 1, 3, and 5. For more information, see Pub. 15-B.



An employer that amends its cafeteria plan to provide a grace period for dependent care assistance may continue to rely on Notice 89-111 by reporting in box 10 of Form W-2 the salary reduction amount elected by the employee for the year for

dependent care assistance (plus any employer matching contributions attributable to dependent care). Also see Notice 2005-42, 2005-23 I.R.B. 1204, available at www.irs.gov/irb/2005-23\_IRB/ar11.html

Box 11-Nonqualified plans. The purpose of box 11 is for the SSA to determine if any part of the amount reported in box 1 or boxes 3 and/or 5 was earned in a prior year. The SSA uses this information to verify that they have properly applied the social security earnings test and paid the correct amount of benefits.

Report distributions to an employee from a nonqualified plan or nongovernmental section 457(b) plan in box 11. Also report these distributions in box 1. Make only one entry in this box. Distributions from governmental section 457(b) plans must be reported on Form 1099-R, not in box 1 of Form W-2.

Under nonqualified plans or nongovernmental 457(b) plans, deferred amounts that are no longer subject to a substantial risk of forfeiture are taxable even if not distributed. Report these amounts in boxes 3 (up to the social security wage base) and 5. Do not report in box 11 deferrals included in boxes 3 and/or 5 and deferrals for current year services (such as those with no risk of forfeiture).

If you made distributions and also are reporting any deferrals in box 3 and/or 5, do not complete box 11. See Pub. 957, Reporting Back Pay and Special Wage Payments to the Social Security Administration, and Form SSA-131, Employer Report of Special Wage Payments, for instructions on reporting these and other kinds of compensation earned in prior years. However, do not file Form SSA-131 if this situation applies and the employee was not 61 years old or more during the tax year for which you are filing Form W-2.

Unlike qualified plans, NQDC plans do not meet the qualification requirements for tax-favored status for this purpose. NQDC plans include those arrangements traditionally viewed as deferring the receipt of current compensation. Accordingly, welfare benefit plans, stock option plans, and plans providing dismissal pay, termination pay, or early retirement pay are not generally NQDC plans

Report distributions from NQDC or section 457 plans to beneficiaries of deceased employees on Form 1099-MISC, not on Form W-2.

Military employers must report military retirement payments on Form 1099-R.

Do not report special wage payments, such as accumulated sick pay or vacation pay, in box 11. For more information on reporting special wage TIP payments, see Pub. 957.

Box 12-Codes. Complete and code this box for all items described below. Note that the codes do not relate to where they should be entered in boxes 12a through 12d on Form W-2. For example, if you are only required to report code D in box 12, you can enter code D and the amount in box 12a of Form W-2. Report in box 12 any items that are listed as codes A through EE. Do not report

General Instructions for Forms W-2 and W-3 (2014)

-16-

			50, 10100.	.,					
			2005	2006	2007	2008	2009	2010	2011
	ME	FICA wage	5.35	5.32	4.78	4.30	4.33	5.56	4.19
		Total wage	10.94	9.41	12.75	8.77	11.33	13.83	9.53
		Minimum	3.18	3.25	3.38	3.50	3.63	3.75	3.75
	NH		4.53	4.70	4.85	4.07	5.47	8.39	5.62
			9.90	10.02	11.48	10.93	16.25	15.22	12.23
			2.38	2.38	2.38	2.38	2.38	3.26	3.26
	VT		3.61	3.21	3.70	4.21	6.99	6.53	6.13
			9.26	7.90	12.37	11.61	14.51	11.90	11.62
			3.65	3.65	3.65	3.72	3.91	3.91	3.95
	MA		7.88	4.85	3.13	5.56	5.41	5.69	3.20
			19.40	10.12	9.49	13.10	12.70	11.18	15.32
			2.63	2.63	2.63	2.63	2.63	2.63	2.63
	RI		4.65	5.03	5.60	4.75	4.80	5.05	4.00
			8.19	10.02	12.86	10.46	10.52	11.55	12.41
			2.89	2.89	2.89	2.89	2.89	2.89	2.89
	СТ		5.71	6.09	8.08	5.44	5.37	6.44	5.29
			14.30	10.11	12.19	12.35	13.50	13.70	11.71
			5.02	5.23	5.41	5.41	5.52	5.69	5.69
	NY		4.56	5.52	6.15	3.70	5.06	6.12	6.33
			12.54	10.31	12.34	9.20	12.18	13.40	11.98
			3.85	4.35	4.60	4.60	4.65	4.90	5.00
	NJ		4.07	8.63	7.33	3.30	8.23	5.96	3.40
			9.79	15.25	15.37	12.40	16.82	12.81	10.96
			3.09	3.09	2.13	2.13	2.13	2.13	2.13
	PA		3.74	3.06	3.71	3.67	3.91	2.24	6.21
			7.75	6.21	7.89	10.12	9.30	7.79	11.35
			2.83	2.83	2.83	2.83	2.83	2.83	2.83
	ОН	YN	3.41	7.93	7.46	3.27	4.67	3.17	3.94
			6.93	12.96	32.17	7.94	10.45	8.55	10.39
			2.13	2.13	3.43	3.50	3.65	3.65	3.70
	IN	$\langle \rangle$	3.10	2.60	3.94	3.98	1.92	2.32	8.05
			6.52	8.77	8.73	9.27	5.43	7.23	17.50
	$\langle \ \rangle$	1	2.13	2.13	2.13	2.13	2.13	2.13	2.13
	IL		6.68	4.39	4.45	4.74	4.69	4.26	6.98
$\sim$			10.15	8.13	11.42	12.01	11.13	11.21	16.91
	)		3.90	3.90	3.90	4.65	4.80	4.80	4.95
	MI		3.17	4.46	8.77	4.19	3.75	3.09	4.93
			6.91	10.32	13.65	8.23	7.80	8.17	8.87
			2.65	2.65	2.65	2.65	2.65	2.65	2.65
	WI		3.47	4.97	3.52	5.32	3.55	4.28	3.24
			9.47	8.97	7.09	8.79	8.06	9.26	9.78
			2.33	2.33	2.33	2.33	2.33	2.33	2.33
	MN		6.12	4.45	5.74	5.27	6.62	6.48	6.20
			12.65	8.73	11.26	9.83	13.16	13.60	12.20
			5.15	6.15	6.15	6.15	6.15	6.15	6.15

Table A1. Average FICA wage, total hourly wage, and the tipped minimum wage by state and year

	2.13	2.13	2.13	2.13	2.13	2.13	2.13
TN	4.45	4.65	4.21	5.11	4.43	4.16	2.57
	7.14	10.40	8.12	10.63	9.41	12.81	10.03
	2.13	2.13	2.13	2.13	2.13	2.13	2.13
AL	2.59	4.29	5.78	4.21	3.61	2.61	1.85
	4.79	8.25	9.93	10.82	5.03	10.72	10.55
	2.13	2.13	2.13	2.13	2.13	2.13	2.13
MS	6.19	3.82	4.27	1.70	4.44	3.42	4.36
	9.94	6.26	11.23	2.83	7.00	7.51	9.48
	2.13	2.13	2.13	2.13	2.13	2.13	2.13
AR	2.28	2.67	6.53	2.46	7.45	3.27	3.79
	7.20	5.28	15.04	6.14	23.56	7.85	8.59
	2.58	2.58	2.58	2.58	3.63	2.63	2.63
LA	6.15	1.47	5.97	3.95	2.57	5.80	5.68
	8.55	5.85	7.62	5.83	9.24	12.82	12.71
	2.13	2.13	2.13	2.13	2.13	2.13	2.13
ОК	2.81	4.39	2.56	6.11	5.50	6.38	5.73
	6.59	12.10	10.01	10.12	8.18	12.19	12.44
	2.58	2.58	2.58	3.28	3.63	3.63	3.63
тх	3.60	4.38	3.22	4.07	4.55	5.49	3.45
	8.17	8.51	7.96	11.38	28.56	10.62	9.39
	2.13	2.13	2.13	2.13	2.13	2.13	2.13
MT	2.01	4.61	6.54	6.35	5.82	5.89	4.88
	2.01	7.77	9.88	8.65	11.09	10.05	6.31
	5.15	5.15	6.15	6.55	7.25	7.25	7.35
ID	3.31	7.23	3.82	3.81	13.16	3.53	3.91
	9.25	10.41	7.70	9.31	28.65	15.51	11.97
	3.35	3.35	3.35	3.35	3.35	3.35	3.35
WY	4.08	2.85	3.14	3.16	7.82	5.90	4.33
NYN	7.41	5.60	6.13	9.63	12.09	9.91	9.75
	2.13	2.13	2.13	2.13	2.13	2.13	2.13
co Co	3.11	2.70	4.40	3.49	4.54	4.13	5.14
	7.82	8.32	11.82	10.74	8.74	12.04	10.97
	2.13	2.13	3.83	4.00	4.26	4.22	4.34
NM	5.93	2.31	5.59	17.22	3.17	2.12	3.76
	9.82	5.75	8.41	22.69	4.58	7.32	11.42
	2.58	2.13	2.13	2.13	2.13	2.13	2.13
AZ	2.26	2.16	3.79	4.16	7.53	7.95	7.74
	7.46	16.50	9.66	12.71	9.09	14.19	17.81
	2.13	2.13	3.75	3.90	4.25	4.25	4.35
UT	2.63	1.77	3.29	5.01	5.31	2.79	3.23
	7.46	5.40	12.85	15.23	13.79	8.35	8.98
	2.13	2.13	2.13	2.13	2.13	2.13	2.13
NV	8.22	8.15	8.24	8.24	10.30	7.74	11.27
	13.08	12.48	12.91	13.48	16.74	16.11	16.88
	5.15	5.15	6.15	6.85	7.55	7.55	8.25
WA	8.56	7.10	7.75	7.89	6.71	7.16	8.25
	TN AL MS AR LA OK TX MT ID WY CO NM AZ	2.13 TN 4.45 7.14 2.13 AL 2.59 4.79 2.13 MS 6.19 9.94 2.13 AR 2.28 7.20 2.58 LA 6.15 8.55 0K 2.81 6.59 2.58 TX 3.60 8.17 2.13 NT 2.01 2.01 5.15 3.31 9.25 3.35 WY 4.08 7.41 2.13 NM 5.93 9.82 2.58 AZ 2.26 7.46 2.58 AZ 2.26 7.46 2.58 AZ 2.26 7.46 2.58 AZ 2.58 AZ 2.26 7.46 2.58 AZ 2.26 7.46 2.13 NV 8.22 13.08 5.15 WA 8.56	2.13       2.13         TN       4.45       4.65         7.14       10.40         2.13       2.13         AL       2.59       4.29         4.79       8.25         2.13       2.13         MS       6.19       3.82         9.94       6.26         2.13       2.13         AR       2.28       2.67         7.20       5.28         2.58       2.58         LA       6.15       1.47         8.55       5.85         2.13       2.13         OK       2.81       4.39         6.59       12.10         2.58       2.58       2.58         TX       3.60       4.38         8.17       8.51       1.13         MT       2.01       7.77         5.15       5.15       5.15         MT       2.01       7.77         MT       2.01       7.77         5.15       5.15       5.15         MT       2.01       7.77         MT       2.01       7.77         MT       2.01       7.77 <tr< td=""><td>2.13       2.13       2.13       2.13         TN       4.45       4.65       4.21         7.14       10.40       8.12         AL       2.59       4.29       5.78         4.79       8.25       9.93         2.13       2.13       2.13       2.13         MS       6.19       3.82       4.27         9.94       6.26       11.23         AR       2.28       2.67       6.53         7.20       5.28       15.04         2.58       2.58       2.58         LA       6.15       1.47       5.97         8.55       5.86       7.62         0K       2.81       4.39       2.56         0.59       12.10       10.01       2.58       2.58         0K       2.81       4.39       2.56         0.13       2.13       2.13       2.13       2.13         0.14       6.01       6.59       12.10       10.01         2.58       2.58       2.58       2.58         10       3.31       7.23       3.82         9.25       10.41       6.54         9.25       10.4</td><td>2.13       2.13       2.13       2.13       2.13         TN       4.45       4.65       4.21       5.11         7.14       10.40       8.12       10.63         2.13       2.13       2.13       2.13       2.13         AL       2.59       4.29       5.78       4.21         4.79       8.25       9.93       10.82         MS       6.19       3.82       4.27       1.70         9.94       6.26       11.23       2.83         AR       2.28       2.67       6.53       2.46         7.20       5.28       15.04       6.14         2.58       2.58       2.58       2.58         AA       6.15       1.47       5.97       3.95         AA       6.15       1.47       5.97       3.95         AA       6.15       1.47       5.97       3.95         CA       2.58       2.58       2.58       2.58         CA       2.58       2.58       2.58       2.66         CA       3.60       4.38       3.22       4.07         S.15       5.15       5.15       6.15       6.55      <tr< td=""><td>213       2.13       2.13       2.13       2.13       2.13         IN       4.45       4.65       4.21       5.11       4.43         2.13       2.13       2.13       2.13       2.13       2.13         AL       2.59       4.29       5.78       4.21       3.61         47.9       8.25       9.93       10.82       5.03         213       2.13       2.13       2.13       2.13       2.13         MS       6.19       3.82       4.27       1.70       4.44         9.94       6.26       11.23       2.83       7.00         213       2.13       2.13       2.13       2.13       2.13         AR       2.28       2.67       6.53       2.46       7.45         7.20       5.28       15.04       6.14       22.56       2.58         ALA       6.15       1.47       5.97       3.95       2.57         8.55       5.85       7.62       5.83       9.24         ACA       4.39       2.266       6.11       5.50         ACA       2.58       2.58       3.28       3.63         AT       8.17       7</td><td>2.13       2.13       2.13       2.13       2.13       2.13       2.13         NN       4.45       4.65       4.21       5.11       4.43       4.16         7.14       10.40       8.12       10.63       9.41       12.81         2.13       2.13       2.13       2.13       2.13       2.13       2.13         AL       2.59       4.29       5.78       4.21       3.61       2.61         4.79       8.25       9.93       10.82       5.03       10.72         MS       6.19       3.82       4.27       1.70       4.44       3.42         9.94       6.26       11.23       2.83       7.00       7.51         AR       2.28       2.57       5.33       2.46       7.45       7.85         AR       2.28       2.58       2.58       3.63       2.63       2.63         LA       6.15       1.47       5.97       3.95       2.57       5.80         AS5       5.85       5.85       7.62       5.83       3.24       12.82         AC       2.81       4.39       2.56       6.11       5.5       5.89         AS5       <t< td=""></t<></td></tr<></td></tr<>	2.13       2.13       2.13       2.13         TN       4.45       4.65       4.21         7.14       10.40       8.12         AL       2.59       4.29       5.78         4.79       8.25       9.93         2.13       2.13       2.13       2.13         MS       6.19       3.82       4.27         9.94       6.26       11.23         AR       2.28       2.67       6.53         7.20       5.28       15.04         2.58       2.58       2.58         LA       6.15       1.47       5.97         8.55       5.86       7.62         0K       2.81       4.39       2.56         0.59       12.10       10.01       2.58       2.58         0K       2.81       4.39       2.56         0.13       2.13       2.13       2.13       2.13         0.14       6.01       6.59       12.10       10.01         2.58       2.58       2.58       2.58         10       3.31       7.23       3.82         9.25       10.41       6.54         9.25       10.4	2.13       2.13       2.13       2.13       2.13         TN       4.45       4.65       4.21       5.11         7.14       10.40       8.12       10.63         2.13       2.13       2.13       2.13       2.13         AL       2.59       4.29       5.78       4.21         4.79       8.25       9.93       10.82         MS       6.19       3.82       4.27       1.70         9.94       6.26       11.23       2.83         AR       2.28       2.67       6.53       2.46         7.20       5.28       15.04       6.14         2.58       2.58       2.58       2.58         AA       6.15       1.47       5.97       3.95         AA       6.15       1.47       5.97       3.95         AA       6.15       1.47       5.97       3.95         CA       2.58       2.58       2.58       2.58         CA       2.58       2.58       2.58       2.66         CA       3.60       4.38       3.22       4.07         S.15       5.15       5.15       6.15       6.55 <tr< td=""><td>213       2.13       2.13       2.13       2.13       2.13         IN       4.45       4.65       4.21       5.11       4.43         2.13       2.13       2.13       2.13       2.13       2.13         AL       2.59       4.29       5.78       4.21       3.61         47.9       8.25       9.93       10.82       5.03         213       2.13       2.13       2.13       2.13       2.13         MS       6.19       3.82       4.27       1.70       4.44         9.94       6.26       11.23       2.83       7.00         213       2.13       2.13       2.13       2.13       2.13         AR       2.28       2.67       6.53       2.46       7.45         7.20       5.28       15.04       6.14       22.56       2.58         ALA       6.15       1.47       5.97       3.95       2.57         8.55       5.85       7.62       5.83       9.24         ACA       4.39       2.266       6.11       5.50         ACA       2.58       2.58       3.28       3.63         AT       8.17       7</td><td>2.13       2.13       2.13       2.13       2.13       2.13       2.13         NN       4.45       4.65       4.21       5.11       4.43       4.16         7.14       10.40       8.12       10.63       9.41       12.81         2.13       2.13       2.13       2.13       2.13       2.13       2.13         AL       2.59       4.29       5.78       4.21       3.61       2.61         4.79       8.25       9.93       10.82       5.03       10.72         MS       6.19       3.82       4.27       1.70       4.44       3.42         9.94       6.26       11.23       2.83       7.00       7.51         AR       2.28       2.57       5.33       2.46       7.45       7.85         AR       2.28       2.58       2.58       3.63       2.63       2.63         LA       6.15       1.47       5.97       3.95       2.57       5.80         AS5       5.85       5.85       7.62       5.83       3.24       12.82         AC       2.81       4.39       2.56       6.11       5.5       5.89         AS5       <t< td=""></t<></td></tr<>	213       2.13       2.13       2.13       2.13       2.13         IN       4.45       4.65       4.21       5.11       4.43         2.13       2.13       2.13       2.13       2.13       2.13         AL       2.59       4.29       5.78       4.21       3.61         47.9       8.25       9.93       10.82       5.03         213       2.13       2.13       2.13       2.13       2.13         MS       6.19       3.82       4.27       1.70       4.44         9.94       6.26       11.23       2.83       7.00         213       2.13       2.13       2.13       2.13       2.13         AR       2.28       2.67       6.53       2.46       7.45         7.20       5.28       15.04       6.14       22.56       2.58         ALA       6.15       1.47       5.97       3.95       2.57         8.55       5.85       7.62       5.83       9.24         ACA       4.39       2.266       6.11       5.50         ACA       2.58       2.58       3.28       3.63         AT       8.17       7	2.13       2.13       2.13       2.13       2.13       2.13       2.13         NN       4.45       4.65       4.21       5.11       4.43       4.16         7.14       10.40       8.12       10.63       9.41       12.81         2.13       2.13       2.13       2.13       2.13       2.13       2.13         AL       2.59       4.29       5.78       4.21       3.61       2.61         4.79       8.25       9.93       10.82       5.03       10.72         MS       6.19       3.82       4.27       1.70       4.44       3.42         9.94       6.26       11.23       2.83       7.00       7.51         AR       2.28       2.57       5.33       2.46       7.45       7.85         AR       2.28       2.58       2.58       3.63       2.63       2.63         LA       6.15       1.47       5.97       3.95       2.57       5.80         AS5       5.85       5.85       7.62       5.83       3.24       12.82         AC       2.81       4.39       2.56       6.11       5.5       5.89         AS5 <t< td=""></t<>

	12.19	10.88	14.88	13.51	11.55	13.08	15.05
	7.35	7.63	7.93	8.07	8.55	8.55	8.67
OR	8.36	6.31	6.62	7.53	8.02	6.29	5.89
	17.42	10.93	13.71	13.90	15.17	12.96	13.17
	7.25	7.50	7.80	7.95	8.40	8.40	8.50
CA	7.23	6.27	6.70	7.96	7.18	7.21	7.63
	14.04	13.67	13.66	14.97	13.31	13.59	14.15
	6.75	6.75	7.50	8.00	8.00	8.00	8.00
AK	6.44	6.80	6.48	9.85	7.88	6.38	7.41
	9.85	12.69	12.34	15.27	14.66	14.75	14.00
	7.15	7.15	7.15	7.15	7.25	7.75	7.75
HI	6.74	6.18	10.94	8.33	9.73	9.27	7.96
	13.11	12.19	19.69	17.10	18.68	15.89	15.73
	6.00	6.50	7.00	7.00	7.00	7.00	7.00

*Source:* Linked CPS ASEC--W-2 file for tax years 2005 to 2011, further linked to the subsample of Outgoing Rotation Group data. Observations remaining in sample have W-2 FICA wages, were found in the rotation group data for the ASEC year and reported hourly earnings, and remained in occupation code 13 (food service). Total observations: 3,636.